Accelerator Status Report

Run Schedule: Jan. 3, 2011 through May 13, 2011
Jan. 19, 2011
Injector    Energy 61.76 MeV
· High current (>100uA) to Hall C is not stable: 0L03 or 0L04 trips. The experts have been investigating. 
· (1/10) High current (160uA) from C laser to the inline dump was set up.
· (1/10) The Wien has been set to -63.2 degrees. 

· Gun conditioning is still in place.

Accelerator   Linac Energy 549.00 MeV
· The optics in the main machine has been stable although the FOPT data showed that the betatron match at Arc 5 changed a bit.
· (1/18) The pre-buncher amplifier failed during the injector beam tuning. 

· PSS key C2 does not work and it has been tagged out. Do not use this key.

· 2L19-7 CM replaced.

· The accelerator has been stable. But the beam energy may be different from that in December. It was reported to the Optics on-call. It may be related to the path length. 

· (1/11) More than a shift was spent to troubleshoot problems with delivery of high current  to Hall C. Experts check the aperture, match, bunch length (the tool does not work), FOPT in the injector. After the injector match was backed out to settings of Jan. 4 and the BLA was calibrated, high current (150uA) to Hall C was established.
· (1/7) During beam delivery to Hall B and beam tuning in Hall C line, the vacuum at the end of the NL deteriorated. It turned out that the beam line was kinked.
· PSS: the injector was dropped twice due to human errors. 

· 30Hz aperture check tool is linked in the BPM menu.

Hall A---
· Will not be ready for beam until Feb. 11 due to installation of a new experiment.
Hall B ---Pass 5, TPE
· nA BPMs had problems on a few occasions, which caused some down time. The ioc was rebooted a few times. The amplifier was replaced a couple of times. (~9 hrs.)
· Beam size is a little bit too large (175 x 208 mm) but the hall has been taking beam.
· Started to take beam on Jan. 6 swing shift.
Hall C--- Pass 1, QWeak 
· High current to the hall is not stable sometimes.

· Compton rate needs to be improved more.

· (1/11) There are problems with high current to Hall C. 

· Started to take beam on Jan. 6 swing shift
